Author's response to reviews

Title: Association between Tumor necrosis factor alpha-238G/A polymorphism and tuberculosis susceptibility: a meta-analysis study

Authors:

Zhijiao Zhang (zhangzhijiao1020@gmail.com)
Hang Zhu (zhuhang86@yahoo.cn)
Xudi Pu (anita10200@hotmail.com)
Fan Zhang (ava11@126.com)
Lei Xun (leixun521@yahoo.com.cn)
Qin Liu (liugin81622@163.com)
Yang Wang (wangyang8289@163.com)

Version: 3 Date: 4 May 2012

Author's response to reviews:

Dear Editors:

Thank you for your reply. The changes we’ve made as you required are as follows: Firstly, we adjusted our article on the basis of MOOSE guideline. Secondly, we add the detailed information about the contribution of each author. Thirdly, we considered we had no competing interests and declared it at the end of the article.

Looking forwards to you!

Best wishes.

Sincerely

Yours,

Zhijiao Zhang